Author's response to reviews

Title: The Effect of Perinatal Anxiety on Bronchiolitis is Influenced by Polymorphisms in ROS-related Genes

Authors:

Eun Lee (unelee@daum.net)
Hyoung Yoon Chang (hyoungyoon@hotmail.com)
Kyung-Sook Lee (debkslee@daum.net)
Dong In Suh (wellcontrolled@hotmail.com)
Ho-Sung Yu (planetho@naver.com)
Mi-Jin Kang (007jokbal@daum.net)
In Ae Choi (inae0920@daum.net)
Jinah Park (pjinah1230@hanmail.net)
Kyung Won Kim (kwkim@yuhs.ac)
Youn Ho Shin (epirubicin13@gmail.com)
Kang Mo Ahn (kmaped@skku.edu)
Ja-Young Kwon (jaykwon@yuhs.ac)
Suk-Joo Choi (drmaxmixchoi@samsung.com)
Kyung-Ju Lee (drkij5094@gmail.com)
Hye-Sung Won (hswon@amc.seoul.kr)
Song I Yang (gitij222@naver.com)
Young-ho Jung (youngdr1@naver.com)
Hyung Young Kim (hawmok77@hanmail.net)
Ju-Hee Seo (juheesea@hanmail.net)
Ji-Won Kwon (pedas@naver.com)
Byoung-Ju Kim (drkbj@hanmail.net)
Hyo-Bin Kim (hbkim1126@naver.com)
So-Yeon Lee (imipenem@hanmail.net)
Eun-Jin Kim (lular1@naver.com)
Joo-Shil Lee (jooshil@korea.kr)
Katherine M. Keyes (kmk2104@cumc.columbia.edu)
Yee-Jin Shin (yjshin@yuhs.ac)
Soo-Jong Hong (sijhong@amc.seoul.kr)

Version: 5 Date: 11 September 2014

Author's response to reviews:

I attached the covering letter with a point by point response to reviewer’s comments.